Epidemic Spreading Model to Characterize Misfolded Proteins Propagation in Aging and Associated Neurodegenerative Disorders



Yasser Iturria-Medina*, Roberto C. Sotero, Paule J. Toussaint, Alan C. Evans* and the Alzheimer's Disease Neuroimaging Initiative¹

Montreal Neurological Institute, Montreal, Quebec, Canada

Abstract

Misfolded proteins (MP) are a key component in aging and associated neurodegenerative disorders. For example, misfolded Amyloid-ß (Aß) and tau proteins are two neuropathogenic hallmarks of Alzheimer's disease. Mechanisms underlying intrabrain MP propagation/deposition remain essentially uncharacterized. Here, is introduced an epidemic spreading model (ESM) for MP dynamics that considers propagation-like interactions between MP agents and the brain's clearance response across the structural connectome. The ESM reproduces advanced Aß deposition patterns in the human brain (explaining 46~56% of the variance in regional Aß loads, in 733 subjects from the ADNI database). Furthermore, this model strongly supports a) the leading role of Aß clearance deficiency and early Aß onset age during Alzheimer's disease progression, b) that effective anatomical distance from Aß outbreak region explains regional Aß arrival time and Aß deposition likelihood, c) the multi-factorial impact of APOE e4 genotype, gender and educational level on lifetime intra-brain Aß propagation, and d) the modulatory impact of Aß propagation history on tau proteins concentrations, supporting the hypothesis of an interrelated pathway between Aß pathophysiology and tauopathy. To our knowledge, the ESM is the first computational model highlighting the direct link between structural brain networks, production/clearance of pathogenic proteins and associated intercellular transfer mechanisms, individual genetic/demographic properties and clinical states in health and disease. In sum, the proposed ESM constitutes a promising framework to clarify intra-brain region to region transference mechanisms associated with aging and neurodegenerative disorders.

Citation: Iturria-Medina Y, Sotero RC, Toussaint PJ, Evans AC, the Alzheimer's Disease Neuroimaging Initiative (2014) Epidemic Spreading Model to Characterize Misfolded Proteins Propagation in Aging and Associated Neurodegenerative Disorders. PLoS Comput Biol 10(11): e1003956. doi:10.1371/journal.pcbi.1003956

Editor: Olaf Sporns, Indiana University, United States of America

Received July 7, 2014; Accepted October 1, 2014; Published November 20, 2014

Copyright: © 2014 Iturria Medina et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability: The authors confirm that all data underlying the findings are fully available without restriction. Data used in this study is available as part of the Alzheimer's Disease Neuroimaging Initiative (ADNI; adni.loni.usc.edu). In addition, this study used the diffusion MRI data of 60 young healthy subjects, from the CMU-60 DSI Template (available at http://www.psy.cmu.edu/~coaxlab/?page_id = 423).

Funding: Data collection and sharing for this project was funded by the Alzheimer's Disease Neuroimaging Initiative (ADNI) (National Institutes of Health Grant U01 AG024904) and DOD ADNI (Department of Defense award number W81XWH-12-2-0012). ADNI is funded by the National Institute on Aging, the National Institute of Biomedical Imaging and Bioengineering, and through generous contributions from the following: Alzheimer's Association; Alzheimer's Drug Discovery Foundation; BioClinica, Inc.; Biogen Idec Inc.; Bristol-Myers Squibb Company; Eisai Inc.; Elan Pharmaceuticals, Inc.; Eli Lilly and Company; F. Hoffmann-La Roche Ltd and its affiliated company Genentech, Inc.; GE Healthcare; Innogenetics, N.V.; IXICO Ltd.; Janssen Alzheimer Immunotherapy Research & Development, LLC.; Johnson & Johnson Pharmaceutical Research & Development LLC.; Medpace, Inc.; Merck & Co., Inc.; Meso Scale Diagnostics, LLC.; NeuroRx Research; Novartis Pharmaceuticals Corporation; Pfizer Inc.; Piramal Imaging; Servier; Synar Inc.; and Takeda Pharmaceutical Company. The Canadian Institutes of Health Research is providing funds to support ADNI clinical sites in Canada. Private sector contributions are facilitated by the Foundation for the National Institutes of Health (www. fnih.org). The grantee organization is the Northern California Institute for Research and Education, and the study is coordinated by the Alzheimer's Disease Cooperative Study at the University of California, San Diego. ADNI data are disseminated by the Laboratory for Neuro Imaging at the University of Southern California. Data used in preparation of this article were obtained from the Alzheimer's Disease Neuroimaging Initiative (ADNI) database (adni.loni.usc.edu). As such, the investigators within the ADNI contributed to the design and implementation of ADNI and/or provided data but did not participate in analysis or writing of this report. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* Email: iturria.medina@gmail.com (YIM); alan@bic.mni.mcgill.ca (ACE)

¶ Membership of the the Alzheimer's Disease Neuroimaging Initiative (ADNI) is provided in Text S2.

Introduction

Misfolded proteins (MP) are associated with aging processes and several human neurodegenerative diseases [1]–[3]. The prion-like hypothesis explains the neurodegenerative progression by the intercellular transfer of pathogenic proteins [4]–[6], under the perspective that MP behave like infectious-like agents that propagate from a few initial host regions to other brain regions. For instance, in Alzheimer's disease (AD), soluble Amyloid-B (sAB) oligomers are thought to be the principal seeds that carry the misfolding process from region to region, accelerating the production/deposition of new misfolded proteins [7]–[9] and thus contributing to drive the pathology to new areas of the brain [10], [11]. The associated A β toxicity has a relevant impact on AD development and progression [12]–[18]. The cell-cell transference is possible because sA β oligomers are very small assemblies of MP, which can be absorbed by axonal processes and transported to cell bodies, causing cytotoxicity in the receiving cells [10], [11], [19]. Also, sA β oligomers that are immersed in the extracellular fluid are subjected to the principles of molecular diffusion processes in the

Author Summary

Misfolded proteins (MP) mechanisms are a characteristic pathogenic feature of most prevalent human neurodegenerative diseases, such as Alzheimer's disease (AD). Characterizing the mechanisms underlying intra-brain MP propagation and deposition still constitutes a major challenge. Here, we hypothesize that these complex mechanisms can be accurately described by epidemic spreading-like interactions between infectious-like agents (MP) and the brain's MP clearance response, which are constrained by the brain's connectional architecture. Consequently, we have developed a stochastic epidemic spreading model (ESM) of MP propagation/deposition that allows for reconstructing individual lifetime histories of intra-brain MP propagation, and the subsequent analysis of factors that promote propagation/deposition (e.g., MP production and clearance). Using 733 individual PET Amyloid-ß (Aß) datasets, we show that ESM explains advanced Aß deposition patterns in healthy and diseased (AD) brains. More importantly, it offers new avenues for our understanding of the mechanisms underlying MP mediated disorders. For instance, the results strongly support the growing body of evidence suggesting the leading role of a reduced A β clearance on AD progression and the modulatory impact of Aß mechanisms on tau proteins concentrations, which could imply a turning point for associated therapeutic mitigation strategies.

brain, i.e. a highly anisotropic movement along the axis of nervous fibers [20]. Consequently, sAß propagation, and the subsequent Aß deposition and cytotoxicity effects, occurs mainly between anatomically interconnected areas or between neighboring neuronal cells [10], [11], [21], [22].

Neuropathologic evidence supports the idea that each neurodegenerative disorder is linked to the misfolding of a specific protein or group of proteins [5], [23]-[25]. Thus, the network degeneration hypothesis proposes that misfolded proteins mechanisms should present disease-specific anatomical patterns [26]-[29]. Two recent studies showed that specific functional and structural covariance subnetworks of the healthy brain are in correspondence with the spatially dissociable cortical atrophy patterns of five distinct dementia syndromes [27], [30]. The reported link between structural/functional brain connectivity patterns and neurodegenerative damage supports the network degeneration hypothesis. This also emphasizes the strategic importance of developing molecular pathological approaches capable of reproducing MP propagation, which might not only be conducive to a better understanding of MP spreading factors, but could also help to evaluate their contribution to disease progression in relation with other postulated pathological mechanisms (e.g. the neuronal activity dependent degeneration [31]-[33]).

In this context, a Network Diffusion Model of disease progression in dementia was proposed [34], where the pathogenic proteins propagation follows the regional concentration gradients under the spatial constraints defined by the brain's connectional anatomy. Consistent with their theoretical predictions, the authors found that specific anatomical sub-modules are in correspondence with characteristic cortical atrophy patterns in AD and behavioral frontal temporal dementia. However, the ability of this model to replicate real MP propagation/deposition patterns remained unexplored. A potential limitation of this model is that it does not consider possible defense mechanisms of the brain. Rather, the disease factors can accumulate gradually, without system resistance, while inducing cellular death and cortical atrophy. Conversely, immunologic brain responses have been demonstrated to combat MP accumulation [35]–[38]. For instance, Aß clearance by macrophages and microglia cells are responsible in part for the remarkable fluctuations in neurological functions that AD patients present even during the same day [26], [35], [39]. Furthermore, recent evidence indicates that initial Aß related processes could have a protective role on the nervous system [40], [41], which suggests non Aß related neurodegenerative effects (e.g. cellular death and cortical atrophy) at all the Aß propagation states but only after an abnormal accumulation process.

Considering the relevance that both intercellular MP transfer and associated clearance defenses have toward the development of neurodegenerative disorders, here we proposed a stochastic epidemic spreading model (ESM) to describe the dynamic interactions between MP infectious-like agents and the brain's clearance response. The validity/applicability of the proposed hypothesis and model was explored using 733 individual PET Aß datasets from the Alzheimer's Disease Neuroimaging Initiative (ADNI). We found that the ESM is able to reconstruct individual/ group AB deposition patterns. Most importantly, ESM predicts that it is not an increased AB production but mainly a deficit in AB clearance processes and an early AB onset age that result in the formation of an excessive AB deposition pattern, and in the conjectured acceleration of the preceding tauopathy. Additionally, our results highlight the strategic role of the MP outbreak regions and their connectional architecture on the disease's temporal progression, as well as the impact of individual genetic and demographic properties on intra-brain Aß propagation.

Results

Recovering the lifetime individual histories of Aß propagation/deposition

We developed a stochastic epidemic spreading model (ESM) to describe intra-brain Aß propagation and deposition processes (*Methods* section). Then, we proceeded to explore the ability of the model to reproduce AB deposition patterns in healthy and pathological brains. Figure 1 illustrates the key processing steps of our approach. First, we used Florbetapir (¹⁸F-AV-45) PET data to quantify AB deposition patterns in a cohort of 733 subjects with non-Hispanic Caucasian ancestry (Table S1) from the ADNI database (Methods, Study participants, Dataset 1). Each participant was previously diagnosed as healthy control (HC, n = 193), early mild cognitive impairment (EMCI, n = 233), late mild cognitive impairment (LMCI, n = 196) or probable AD (n = 111). For each subject, the baseline ¹⁸F-AV-45 PET scan was employed to calculate the AB deposition probabilities for 78 regions covering all the gray matter [42], and these were used to define the individual AB deposition pattern (Methods, Regional $A\beta$ deposition patterns). Next, we used the developed ESM, and region-region anatomical connectivity information from 60 healthy young subjects (Methods, Study participants, Dataset 2), to generate multiple hypothetical regional courses of AB propagation/deposition. Each hypothetical generation corresponded to a specific set of sAB spreading seed regions, up to a maximum of 6 regions consisting of all possible combinations, and a set of model parameters from which we simulated 50 years of propagation starting with AB presence only in the seeds. A selective iterative algorithm (Methods, Model exploration/validation) was used to identify the seed regions that better explained the PET-based Aß deposition patterns across the study cohort, as well as the individualized model parameters that maximized the similarity between the generated and the individual reference AB deposition patterns. In sum, a set of the most likely



Figure 1. Reconstruction of individual Aß propagation/deposition histories using an ESM. ¹⁸F-AV-45 PET scans (**A**) are used to calculate individual Aß deposition patterns for different regions covering all the brain's gray matter (**B**). Then, detailed region-region anatomical connectivity information from a young healthy group (**C**) is used to generate multiple hypothetical lifetime Aß propagation/deposition courses (**D**). Each hypothetical course corresponds to an initial set *i* of sAß spreading seed regions and a different set of global model parameters $\theta_i = [\beta_i, \delta_i, \sigma_i]$. Then, a selective iterative algorithm estimated, for each subject, the model parameters that maximized the similarity between the generated and the reference Aß deposition pattern, as well as the time point at which this maximization occurred. The latter output was used to calculate the individual onset age of Aß binding, which in conjunction with the obtained model parameters were assumed to characterize each subject's Aß propagation/deposition history.

doi:10.1371/journal.pcbi.1003956.g001

Aß outbreak regions were identified, assuming the same set of regions for the whole sample, whereas for each subject four different model parameters were estimated: Aß production rate (β), Aß clearance rate (δ), onset age of Aß outbreak (Age_{onset}), and model noise level (σ). For further details see *Methods (Model exploration/validation* subsection) and Figure 1.

Consistent with the hypothesis of an intra-brain AB epidemic spreading behavior, our propagation/deposition model reproduced, from the remote non-binding states, the characteristic AB deposition patterns in the adult cohort (Figures 2A,B). It explained between $46.4 \sim 56.8\%$ (all $P < 10^{-10}$) of the variance in mean regional AB deposition probabilities (adjusted by age, gender, and educational level) in HC, EMCI, LMCI and AD groups. See Table S2 for a comparison with previous approaches. In addition, it identified the posterior and anterior cingulate cortices as the most probable starting seed regions for the AB propagation process (see Table S3 for examples of other tested combinations of regions, based on previous reports). The cingulate cortex, particularly its posterior area, is considered a core node of the default mode network (DMN), and is thought to be involved in self-relevant/ affective decisions, mental simulation, and integration tasks [43], [44]. This result is in agreement with the large amount of evidence suggesting the critical role of the DMN on the genesis and propagation of AD [31], [45], [46]. For a complementary seeds identification analysis, see *Discussion section (Identification of the MP propagation epicenter* subsection) and Figure S1.

Next, we re-evaluated the competence of the ESM framework to reproduce prion-like spreading mechanisms, but now based on the idea that, if the ESM is describing real intra-brain propagation of MP, then alterations in the structural connectional information should affect the model's results negatively. We tested this by comparing the capability of the ESM to explain advanced AB deposition patterns, using the available connectivity information and alternatively using "non-informative" connectional information. For this, 100 randomized versions of the original anatomical connectivity matrix were created (preserving its weight, degree and strength distributions [47]), and the propagation model was evaluated for each of these versions. We observed a significantly higher model competence (all $P < 10^{-5}$) to explain the AB deposition patterns when the original anatomical connectional information was used (Table S4). This result supports the ability of the SEM to describe real MP spreading processes, based on the central interrelation between biological factors directly related to these pathogenic proteins (e.g. Aß production and clearance) and the complex connectional architecture of the human brain.



Figure 2. Characteristic regional Aß deposition patterns in healthy and pathologic brains. A) PET-based mean regional Aß deposition probabilities (adjusted by age, gender, and educational level) in HC, EMCI, LMCI and AD groups. Nodes correspond to 78 regions covering all the brain's gray matter, with node sizes proportional to the associated Aß burden. Note the progressive expansion of the Aß deposition, starting mainly from DMN regions to the rest of the brain. This supports the development of an abnormal Aß deposition pattern in correspondence with the disease progression (from HC to advanced AD clinical states). B) Correspondence between the estimated and PET-based mean regional Aß deposition probabilities for the different groups. C) Prediction accuracy distributions obtained for the different groups (via a repeated random sub-sampling cross-validation procedure). doi:10.1371/journal.pcbi.1003956.g002

Finally, the statistical robustness and predictive power of the introduced ESM was tested via a repeated random sub-sampling cross-validation. Each clinical group (HC, EMCI, LMCI and AD) was randomly split into training and test data of the same size. For each such split, the model values derived at the group level for the training data were used to test the predictive validity of the model on the validation group. We observed significant predictive power across the different clinical states (Figure 2C), with prediction accuracy values ranging from 40.7% (95% CI: 36.3, 45.0) for the HC group to 31.4 (95% CI: 28.4, 34.2) for the AD group (Table S5). Slightly lower prediction accuracy was observed for the AD group. We attribute this to the smaller sample size, in comparison with the other groups, and as will be analyzed in the next subsections, to a larger period of Aß propagation/deposition processes (with a significantly earlier

propagation onset). This larger period of the phenomenon to be modeled can be consequently associated to a larger accumulation of model errors.

Predicting regional Aß arrival time with effective anatomical distance to outbreak region and connectivity degree

Historically, the identification of outbreak nodes has been considered a primary step towards the spatiotemporal understanding of epidemic phenomena [48]. In the context of brain neurodegenerative disorders, functional proximity to epicenter regions implies greater disease-related regional vulnerability [30]. This suggests an organized pattern for propagation of disease agents, in accordance with the trans-neuronal network-based MP spread hypothesis, and supports the key role of specific epicenter regions in the disease progression processes. Those results were obtained using an indirect measure of MP presence, i.e. gray matter atrophy quantified using voxel-based morphometry. However, the relation between gray matter atrophy and MP effects is still unclear, and, in addition, the former can also be caused by multiple different factors (e.g., vascular dysregulation).

To obtain direct evidence of MP dispersion as a function of proximity to an epicenter, we first explored the relation between the PET-based regional AB deposition patterns and the effective anatomical distances to the identified AB outbreak regions (anterior and posterior cingulate cortices; Methods, Model exploration/validation). We observed a significant negative linear relationship between these two variables, across the four clinical states (Figure 3A). Interestingly, best-fit lines for the different clinical groups displayed a consistent co-linearity (Figure 3A and Figure S2). The relationships were characterized by similar slope but different AB deposition intercepts that increase according to disease progression. We verified that these associations are not explainable by the spatial proximity between regions (Table S6). These results support the role of the outbreak regions as centers of radial disease factor propagation, which is modulated by the brain's connectional architecture.

Next, we used the spatiotemporal information provided by the ESM to analyze the link between regional AB arriving times $(T^{A\beta})$ and the effective anatomical distances. For each brain region i, $T_i^{A\beta}$ was calculated as the time at which the AB probability deposition of this region reached a given threshold (e.g. $P_i < 0.9$ implying no deposition, $P_i \geq 0.9$ implying deposition). In line with the previous results, we found a significant linear predictive relationship between the effective anatomical distances and the $T^{A\beta}$ values (Figure 3B). The shape of this relationship was invariant to the selection of different AB deposition thresholds.

Notably, these results correspond with the linear predictive relationship reported for effective distances in human social networks and disease arrival times for real epidemics propagation data [49] (e.g. 2009 H1N1 pandemic). This parallelism between intra-brain MP propagation mechanisms and epidemic propagation in human disease networks [49], supports our hypothesis of an intra-brain epidemic spreading behavior of MP propagation. Furthermore, these model-based findings clarify the distance-vulnerability effects observed for gray matter atrophy [30] and Aß deposition (Figure 3A).

In terms of regional vulnerability to disease pathological effects, recent studies have also suggested a direct link between structural/ functionally connectivity levels and regional vulnerabilities [31], [50], [51]. Highly connected brain regions are usually known as "hubs nodes" of the brain network (for review see [50]). Buckner et al., 2009, showed a high correspondence between AB deposition levels and functional connectivity in the brain hubs. Further evidence, based on meta-analyses of published magnetic resonance imaging data about 26 different brain disorders, suggest that pathological brain lesions (i.e. gray matter atrophy lesions) are mainly concentrated in structural hub regions, independently of the studied disorder [51]. This fact is considered a consequence of the high topological centrality and biological cost of the hubs, which make them more vulnerable to a diverse range of pathogenic processes [31], [51]. In order to explore if the introduced ESM can clarify this connectional-pathogenic association, we analysed the relation between regional anatomical connectivity degrees and AB arrival times, as measures of "hubness" (Methods, Anatomical connection probability) and temporal vulnerability to receive aberrant disease factors, respectively. We observed significant negative correlations (all $P < 10^{-9}$)

between these two variables, independently of the selection of different A β deposition thresholds (see Figure S3). This suggests that regions with higher anatomical connectivity degrees experience early A β arrival and, consequently, larger periods of exposition to the negative effects of this aberrant protein.

A malfunctioning Aß clearance system and an early Aß onset age are major factors associated with AD

For decades, AB propagation and accumulation has been thought to have a causal role on the cascade of cognitive/ clinical events leading to AD [52], [53]. For instance, AB toxicity has been causally associated with brain oxidative stress [14], [18], mitochondrial dysfunction [18], synapse and spine loss [13], widespread neuronal dysfunction and cell death [12], synaptic plasticity and memory impairment [16], [17]. To test the potential clinical impact that progressive AB presence can have on the pathology's progression, we studied whether model variables controlling intra-brain Aß propagation/deposition are related to AD and intermediate cognitive/clinical states. For this, we considered the clinical diagnosis (HC, EMCI, LMCI or AD) as a dependent variable in a Multinomial Logistic Regression model with AB production/clearance rates, noise and onset age as independent variables (controlling by gender, age and educational level). Then, the contribution of each regressor was evaluated using a robust metric of relative importance in prediction analysis (Methods, Statistical Analysis).

We observed a statistically significant relationship between clinical diagnosis and AB clearance rate and onset age (Figure 4 and Table S7). The clearance rate was found more related to the clinical diagnosis than the other model parameters (Figure 4A), explaining 8.45% (95% CI: 4.88, 11.89) of its inter-subject variance. A closer look at the differences between the four clinical groups (Figures 4B-E and Table S8), revealed that the clearance rate is also the model variable with the most consistent variance across the different clinical diagnoses. It is followed by the onset age of AB accumulation, which reflects a decreasing transition from HC to EMCI-AD states (Figure 4D) and explains 6.77% (95% CI: 3.42, 9.88) of the variance in clinical diagnoses. We observed a significant decreasing trend on the AB production rate from HC to EMCI-AD states (Figure 4B), however the impact of this effect on the clinical diagnosis was not significant (95% CI: -0.76, 3.77; for further analysis, see Text S1). While the individualized global AB production rate can be seen as a measure of the lifetime individual regional potential to produce Aß infectious-like factors, the corresponding clearance rate reflects the lifetime intrinsic capacity to combat the AB accumulation. Therefore, these results suggest that although a significantly earlier onset age of AB accumulation and a nonsignificantly lower production rate of AB agents are associated to AD, a deficiency associated to AB clearance might be the most determining AB-mediated factor for the development of the disease.

APOE e4 genotype multi-factorial impact on Aß propagation/deposition

Apolipoprotein E (APOE) e4 genotype is considered a relevant genetic risk factor for AD and intermediate MCI states [54]. It has been associated to A β aggregation into fibrils [55], the hindered clearance of sA β [56], and neurodegeneration by directing toxic A β oligomers to synapses [54], [57]. Using our ESM, we explored how different APOE e4 genotypes impact A β propagation and deposition. For each model parameter, we performed a three-way



Figure 3. Effective anatomical distance to outbreak regions modulates the Aß propagation processes. A) PET-based regional Aß deposition probabilities for the different groups vs effective anatomical distances. B) Regional Aß arriving times vs effective anatomical distances, for different Aß probability thresholds (i.e. 0.1, 0.5 and 0.9). In A) and B), note the co-linearity between different clinical states or Aß probability thresholds, with more advanced disease states corresponding to higher depositions and propagation times. See also Figure S2. doi:10.1371/journal.pcbi.1003956.g003

ANOVA test considering as grouping parameters the number of APOE e4 allele copies, as well as the gender and educational level of the participants.

We observed a significant effect of APOE e4 genotype on the Aß production/clearance rates and on the onset age (Figure 5A and Table S9). In particular, we found that APOE e4 genotype



Figure 4. Subjects with different clinical states presented different Aß propagation histories. A) Explained variance of the clinical diagnoses (HC, EMCI, LMCI and AD) by the different Aß propagation model parameters. Mean (\pm standard error) Aß production rate (**B**), Aß clearance rate (**C**), noise standard deviation (**D**) and onset age of Aß propagation (**E**) for the different clinical groups (adjusted for gender and educational level). *p<0.05, **p<0.01, ***p<10⁻⁵, Student's *t*-test. doi:10.1371/journal.pcbi.1003956.g004

had highest impact on the onset age, decreasing it proportionally to the number of APOE e4 allele copies (Figures 5A,E), and explaining 13.21% of its inter-subject variance ($P = 1.12 \times 10^{-24}$, F = 59.57). This result is in line with previous reports associating APOE e4 genotype with an earlier age at disease onset and a faster AD pathological progression [58], [59]. In addition, we observed a significant decrease in AB clearance rate with regard to the number of APOE e4 allele copies (Figures 5A,C), explaining 10.48% of its inter-subject variance ($P = 2.24 \times 10^{-19}$, F = 45.60). This supports our previous result associating AD onset with an AB clearance deficiency and, more importantly, evidences that this clearance deficiency partly has a genetic cause [56]. We also found significant effects of APOE e4 on AB production rate (Figures 5A,B; $P = 5.38 \times 10^{-19}$, F = 21.98), which reflects the multifactorial influence of this genotype on the evolution of AD and intermediate MCI states [54]-[57].

Further statistical analyses were performed to assess how the specific number of APOE e4 allele copies impact on Aß propagation and deposition (Table S10). We found that the effects due to the presence of two e4 allele copies are more relevant (in terms of the model parameters) than the effects due to the presence of only one copy (Figures 5B–E and Table S10). This is in agreement with the reported semi dominant inheritance effect of APOE genotype on developing AD [60].

When investigating the relationship of the model parameters with the demographic variables, we also found a significant impact of gender on AB production rate ($P = 1.90 \times 10^{-3}$, F = 9.68), AB clearance rate ($P = 1.35 \times 10^{-3}$, F = 19.19) and AB onset age ($P = 2.06 \times 10^{-3}$, F = 36.84) (Tables S9 and S11). For all these cases, female gender was associated with significantly lower model

parameter values (Table S11). This result is in high correspondence with the fact that women are more likely to develop AD than men [61], [62]. Furthermore, we found a significant interaction between APOE e4 genotype and gender, which together are modulating the AB onset age ($P=1\times10^{-5}$, F=9.30). This is consistent with the higher propensity for women to develop AD across most ages and APOE genotypes [62], with the most pronounced detrimental effect of APOE e4 on DMN connectivity and CSF tau levels [61], and with the reported greatest amyloid plaque and neurofibrillary tangle pathology for women [63].

Finally, when investigating the relationship of the noise parameter σ with APOE e4 and the demographic variables (Table S9) we found that female subjects with a higher educational level have a higher noise level (P = 0.019, F = 5.45). In conjunction with a significant impact of gender and educational level on the AB onset age (P = 0.01, F = 5.45) and a non-significant trend effect of educational level on AB clearance rate (P = 0.093, F = 2.82), this may be reflecting the complex relationship that exists between AB binding, gender, cognitive reserve and clinical state [64]. The larger variability in AB deposition patterns associated with higher noise, gender and educational level, could explain the disputed results of the cognitive reserve hypothesis [65]–[67].

Modulatory impact of AB propagation/deposition history on CSF AB^{1-42} , t-tau and p-tau levels

CSF measures of AB, total tau (t-tau) and phosphorylated tau (p-tau) are considered the most relevant early biomarkers of AD [68], [69]. Although AB and tau proteins were historically considered to



Figure 5. Multi-factorial impact of APOE e4 genotype on Aß propagation/deposition. A) Explained variance of the propagation model parameters by the different APOE e4 genotypes (zero, one or two e4 allele copies). Mean (\pm standard error) Aß production rate (**B**), Aß clearance rate (**C**), noise standard deviation (**D**) and onset age of Aß propagation (**E**) for the different number of e4 allele copies (adjusted for gender and educational level). *p<0.05, **p<0.01, ***p<10⁻¹⁰, Student's t-test. doi:10.1371/journal.pcbi.1003956.g005

arise and act independently, now it is thought that both proteins are strongly interrelated [13]. Based on different converging evidences, it has been suggested that Aß pathophysiology might drive and accelerate pre-existing tauopathy [70]. Here, we aimed to re-evaluate this interrelation hypothesis under the assumption that, if the intra-brain ESM of Aß propagation/deposition can reflect Aß pathophysiology accurately, then abnormalities in CSF Aß, t-tau and p-tau concentrations should be correctly reflected by the individualized model parameters. For this, we used CSF Aß¹⁻ ⁴², t-tau and p-tau¹⁸¹ measures from a subsample of 307 healthy and non-healthy subjects belonging to the ¹⁸F-AV-45 PET scanned group (*Methods, CSF measures*). For each CSF measure, we performed a seven-way ANOVA test, considering the model parameters, age, sex and educational level as modulatory factors.

The results (Figure 6A and Table S12) show a significant impact of AB production/clearance rates on CSF AB^{1-42} , explaining $(P = 1.24 \times 10^{-12},$ 10.40% F = 55.02) and 11.85% $(P = 4.83 \times 10^{-14})$, F = 62.66) respectively, of its across-subject variance (see also Text S1). We also found that the AB onset age and the chronological age are significant modulators of CSF $A\beta^1$ ⁴², explaining 2.31% ($P = 5.36 \times 10^{-4}$, F = 12.24) and 2.97% $(P = 9.29 \times 10^{-5}, F = 15.69)$ respectively, of its variance. Together, all considered modulators accounted for 28.82% of the CSF AB¹⁻ ⁴² variance. Aß production/clearance rates were also found to have significant impact on CSF t-tau, explaining 4.45% $(P = 1.68 \times 10^{-5})$, F = 19.33 and 2.77% $(P = 6.32 \times 10^{-4})$, F = 11.93) respectively of its variance. However, in this case the higher impacts correspond to the AB onset age and chronological age (Figure 6B), with 5.08% (P=4.41×10⁻⁶, F=21.87) and 5.45% (P=2.07×10⁻⁶, F=23.44) respectively, of explained variances. Similar effects were observed for CSF p-tau (Figure 6C), for which AB onset age was the strongest modulator and accounted for 4.44% (P= 2.07×10^{-5} , F=18.23) of its variance.

According to these results, while Aß production/clearance rates might be influencing the deposition and recirculation of Aß and subsequently its inter-relationship with tau proteins, the observed Aß onset age and chronological age effects on t-tau and p-tau may be reflecting the time duration of such inter-relationship. These results are consistent with the idea of an interrelated pathway between amyloid pathophysiology and tauopathy [70], [71] and, in combination with results from the previous subsections, they are also consistent with the notion of an associated failure to clear mislfolded proteins [70], [72].

Discussion

Characterizing the mechanisms underlying intra-brain MP propagation and deposition constitutes a major challenge of the molecular pathological approaches devoted to the study of neurodegenerative disorders. Here, we showed that these complex mechanisms can be biophysically described by epidemic spreading-like interactions between the infectious-like agents (misfolded proteins) and the brain's clearance response, across the human structural connectome. We identified several genetic, structural and demographic factors associated to the biophysical model variables controlling these interactions. The proposed ESM



Figure 6. Influence of parameters controlling Aß propagation/deposition on CSF A B^{1-42} , t-tau and p-tau¹⁸¹ levels. While CSF A B^{1-42} (A) is mainly influenced by Aß production/clearance rates, t-tau (B) and p-tau (C) are highly influenced by chronological age and Aß onset age (combined, these two temporal factors should reflect the interrelation period between amyloid pathophysiology and tauopathy). doi:10.1371/journal.pcbi.1003956.g006

constitutes a promising framework to clarify intra-brain region to region transference mechanisms related to aging and neurodegenerative disorders.

The prion-like hypothesis and the misfolded proteins epidemic spreading behavior

The prion-like hypothesis explains the neurodegenerative progression by the intercellular transfer of pathogenic factors [4], [73]. This perspective presents a striking similarity with the spread of real infectious diseases in biological populations. Social networks constitute a common structural substrate over which infectious factors propagate, reaching in some cases an epidemic/ uncontrollable behavior [74]. Independently of the pathogenic agent's characteristics, its propagation dynamics are always constrained by the connectivity structure of the attacked system. It is in this context that we hypothesized the AB proteins propagation and deposition as a natural epidemic spreading event, whose dynamics are determined by infectious-like agents and immunologic response actions that compete under a restrictive anatomical network (the structural human connectome). Note, however, that the term infectious does not necessarily imply the presence of fully negative propagating factors, since the genesis and role of MP in the brain are not completely understood [40].

Previous studies have used the brain's structural and functional connectivity to explain neurodegenerative atrophy patterns (for recent reviews see [75], [76]). We extended previous connectivity-based approaches [27], [30], [34] by combining pathogenic factors actions (production and spreading) with possible defense responses, including also the influence of stochastic or undetermined processes. The inclusion of basic biological variables (e.g. MP production/clearance rates, time of propagation) provides a more realistic characterization and understanding of the studied phenomenon, allowing not only to reproduce the MP dynamics but also to identify the genetic, structural, and demographic factors associated to it. For purposes of comparing different methods, we applied the Network Diffusion Model (NDM) [34] to the same AB datasets and connectivity information (for further details see Table S2). We found that NDM also identified the posterior and anterior cingulate cortices as the most probable starting seed regions for the Aß propagation process. However, even when the obtained mean regional explained variance for the NDM was around 27-33%, with a significant statistical association (p < 0.05), the corresponding root mean square errors (RMSEs) were considerably high, reflecting large absolute differences between estimated and reference AB concentration patterns. In addition, Akaike Information Criterion (AIC) values evaluated for both models (ESM and NDM) revealed a significantly lower accuracy performance for the NDM ($P = 7.13 \times 10^{-8}$, Z = -5.26), independently of the number of models parameters. We noted that although the NDM is capable of dispersing the initial infectiouslike factors from the seed regions to the rest of the brain network, such dispersion is at the expense of the local concentrations, which after the initial exchange decreases continuously. As a consequence, the total AB concentration is never higher than the "injected" amount and after a given time the propagation of the factors stops. This behavior is not physiologically realistic as shown in the literature [77]. Note that this issue is a consequence of the absence of a source term in the NDM, which is included in the ESM.

In addition, consistent with reported associations between functional proximities to a pathogenic epicenter and gray matter atrophy levels [27], [30], we found that effective anatomical distances to the AB outbreak regions can predict regional AB depositions and arrival times values. In terms of prediction accuracy, anatomical connectional proximities to the epicenter seem to be more interrelated to AB levels than functional proximities to gray matter atrophy levels (Table S2). This might be responding to several possible causes, such as: a) a tentative higher impact of the anatomical connectivity (implying only direct links) than the functional connectivity (implying both direct and indirect links) on pathogenic agents propagation, b) the use on [27], [30] of indirect measures of MP presence to evaluate prion-like mechanisms, i.e. gray matter atrophy quantified using voxel-based morphometry, and c) the fact that gray matter atrophy can be caused by multiple pathogenic factors (e.g., vascular and metabolic dysregulations). In addition, in these studies the nodes of the analyzed networks were obtained based on a priori statistical selection of the significantly affected brain regions in the diseased group, ignoring other brain regions, which may have introduced a bias in the posterior atrophy level vs functional proximity analysis.

Anatomic connectivity impact on intra-brain MP propagation

The intercellular transference of pathogenic proteins (e.g. across axonal projections [10], [19], or the extracellular space that is constrained by the connectional architecture [20]), is a major statement of the prion-like hypothesis. The ability of ESM to reconstruct AB deposition patterns from early to advanced disease states, suggest the methodological importance of considering the structural connectional information on the modeling of MP propagation/deposition mechanisms. However, these alone do not offer an evaluation of the real contributions/advantages of using or not the connectional information on MP propagation modeling. We tested this contribution by comparing the ability of the introduced ESM framework to explain advanced AB deposition patterns, using the available connectivity information and alternatively using equivalent non-informative connectional information (Results, first subsection, and Table S4). The results supported a significantly higher model competence to explain AB deposition patterns when the original anatomical connectional information was used. Similarly, effective anatomical distances estimated using the structural human connectome were significantly better predictors of regional AB levels (all $P < 10^{-11}$) than the equivalent distances estimated using the randomized networks. In addition, we verified that spatial proximities of the considered 78 regions to the identified epicenter (cingulate cortex) cannot explain the observed predictive relationship between effective anatomical distances to epicenter and regional AB levels (Table S6). Together, these results add evidence to the previously postulated prion-like mechanisms associated to AB proteins [10], [19], [39] and, furthermore, highlight the methodological importance of considering structural brain connectivity information for the study of such processes.

In accordance with previous studies [31], [32], [50], [51], our results support a significant relationship between regional connectional degrees and pathological vulnerability. We observed that regions with higher connectivity degrees are primarily targeted by the AB factors (Figure S3). This suggests that hub regions in the brain are most extensively exposed to the negative effects of these aberrant proteins. Considering that similar connectivity-based modulatory effects could be presented for different MP (e.g. tau, α -synuclein) or even, as previous literature is suggesting, for other transferable region-region pathogenic mechanisms (e.g. metabolic or functional dysregulations), this phenomenological relationship could explain the spatiotemporal association between regional hubness and lesional levels observed for different diseases.

Identification of MP propagation epicenters

Converging evidence suggests that cingulate cortex is one of the earlier structures presenting AB deposition [29], [31] and other structural/functional alterations related to AD progression [31], [46], [51]. Therefore it is not surprising to find this structure as a strong candidate to be the AB outbreak region and to have a consistently high Aß deposition across the whole sample. However, with the current data it is hard to determine whether the identification of this structure, or any other, as propagator epicenter is more related to: a) the fact that it is the real AB outbreak center, or b) it is located spatially close to the real outbreak, or c) its selection is reflecting a model limitation (e.g. potentially, the model could not identify peripheral regions but only some kind of "best MP propagator"). Despite this intrinsic limitation, we know that while a realistic selection should be influenced by the deposition level of each seed candidate, it should also depend on the seed's anatomical connectivity patterns. In order to explore this and its relation with our previous results, we performed a complementary analysis for the seed identification, now assuming that the regional connectional proximities to the real outbreak should be capable to reflect the regional levels of AB deposition, as a direct consequence of an aberrant factor propagation from the initial center [49]. For this, we calculated the correlation between the regional AB deposition values and the effective anatomical distances to each considered brain region. The results (Figure S1) suggested that, in accordance with the previous reasoning, the cingulate regions are the most likely candidates to have a leading role in AB outbreak and subsequent propagation, presenting the highest correspondences between connectional distances and brain Aß deposition levels. However, a conclusive validation of these results requires more integrated data, e.g. longitudinal PET AB datasets from initial to more advanced AB binding states, and/or animal models to evaluate the level of mismatches between a given MP/AB injection site and the corresponding identified epicenter.

Aß clearance deficiency and implications for medication therapies

Historically, $A\beta$ binding features associated with AD have been assumed to be a causal consequence of the imbalance between $A\beta$ production and clearance. With the exception of rare genetic forms of AD [78], an increased AB production in typical AD patients has not been consistently confirmed [38], [79]. In fact, recent metabolic labeling analyses in a cohort of AD (n = 12) and normal controls (n = 12) revealed a non-significant inter-group difference in the A β production rates, while clearance rates were found significantly decreased in the AD group [38]. Moreover, new evidence suggests that under certain conditions AB proteins may play a protective role on the nervous system [40], [41]. Our results are in agreement with a decreased AB clearance capacity associated with AD onset [38], [39]. Once we evaluated the ESM variables associated to AB propagation/deposition with real data, we found a significant association between individual AB clearance rates and clinical states. In terms of the relative importance as a clinical predictor, the clearance rate demonstrated a considerably higher impact in comparison with the other variables (i.e. Aß onset time and production, physiological/external noise). This result strongly suggests that differences in the capacity to clear/degrade the AB proteins result in vast differences in the net amounts ending up in the receiving cells [39], which can cause more seeding and pathology in these cells. At the cellular level, ubiquitin-proteasome and the autophagy-lysosome pathways are considered the two main routes for intracellular MP protein degradation [72]. Although the ESM cannot distinguish between these two pathways, the obtained results are in some way highlighting the importance of considering these and other similar mechanisms related to MP clearance, which might have a relevant role on the development and progression of different neurodegenerative diseases. Specifically, the growing body of evidence supporting a reduced A β clearance in AD development could imply a turning point for associated therapeutic mitigation strategies [35], [39], [72]. A significant increment of $A\beta$ and tau clearance capacities attending in parallel to the individual genotype and demographic characteristics, through an immunologic reinforcement [35], [80], or through a genetically induced enhancement [81], may be, tentatively, an alternative to combat AD onset and progression, with the subsequent impact on the associated undesirable symptoms.

A β and other factors contributing to neurodegeneration

Although ESM was created to describe the spatiotemporal propagation of different MP (AB, tau, $\alpha\text{-synuclein},$ superoxide

dismutades-1, etc), here it was explored only with Aß patterns. As it is well known, Aß is not the unique pathogenic factor associated with neurodegenerative progression. In AD and intermediate mild cognitive states, tau proteins are thought to also have a primary role on disease progression, presenting a higher correspondence than Aß to gray matter atrophy patterns and clinical states [70]. However, Aß toxicity has been evidenced in different forms and its negative role is currently the subject of scientific debate. Aß toxicity has been causally related to brain oxidative stress [14], [18], mitochondrial dysfunction [18], synapse and spine loss [13], widespread neuronal dysfunction and cell death [12], synaptic plasticity and memory impairment [16], [17]. Moreover, as evidenced by previous studies [13]–[15] and as was verified in our analysis (Results, Figure 6), Aß proteins have a significant modulatory impact on tau proteins concentrations.

A hierarchical model of AB acting on tau is supported by several lines of evidence [13]. As a consequence, many scientific groups are now emphasizing the strategic importance of considering the mutual interrelation (AB, tau) as an alternative to get a better understanding of the pathogenic mechanisms and clinical consequences associated to both proteins [80]. Furthermore, the current consensus is that a given clinical state can not only be caused independently by one protein (AB or tau), but it can also derive from complex interactions between these and/or other contributing factors (e.g. metabolic, vascular and functional dysregulations).

Methodological issues and future work

This study presents several limitations. The anatomical connectivity information used in the model evaluation still presents notable limitations [75], [82]. In this sense, current difficulties to distinguish between real and spurious connections may have a significant impact on the results, by distorting the modeled structural relations among the regions and consequently the probabilities of receiving or not new MP agents from connected regions. The symmetrical property of the connectivity matrix, which is the result of the limitations of current diffusion MRI tractography techniques, makes it impossible to consider anterograde or retrograde MP propagation processes, which could potentially present different kinetic mechanisms. Additionally, the use of a gray matter parcellation template with variably sized regions [42], may introduce a significant bias in the connectivity evaluation. The probabilistic connectivity measure used in this study is based on regional maximum-voxel levels of connectivity, and thus reflects the fibers orientational coherence across the estimated connecting fibers pathways [83]. This approach may be less sensitive to the sizes of the regions than other traditional probabilistic connectivity measures, e.g. the frequentist ratio between the number of connecting and generated fiber pathways, with different seed sizes implying different number of generated paths. However, independently of the connectivity approach used, it is still not clear how to evade the influence of the nodes selection on the structural network's estimation [84]. While model estimation at the voxel level could be a tentative solution to this issue, such approach would imply other limitations, such as low inter-subjects correspondence across nodes, the difficulty to interpret regional findings, and a considerably high computational cost. In the ESM, regional MP production and clearance rates depend of the local MP deposition probabilities, which implies different effective rates across the time and brain regions. But, at the same time, these local effective rates are subjected to global individual parameters of MP production and clearance (β_o and δ_o in Equations 4 and 5, respectively). This generalization may cause the loss of real variability at the local tissue levels, such as possible spatial differences in MP production/clearance rates due to changes in neuronal and glial cells properties across the cortex. However, the inclusion of regional production/clearance parameters in the model could lead to a very high dimensional parameters space (i.e. with at least two additional parameters per region), and subsequently result in considerably higher difficulty to evaluate these at individual levels. In addition to these limitations, errors in defining an appropriate gray matter parcellation scheme [85] and the characteristic low signal to noise ratio of the 18F-AV-45 PET data are factors that may affect the global capability of the proposed ESM to predicts regional A β deposition patterns.

This study can be extended in multiple directions. These should include further validation of the developed formulation with animal models and longitudinal datasets, in order to characterize its robustness and predictive competence. In this sense, it would be of relevance to validate the identified epicenter regions, and to evaluate the ability of the model to predict future MP deposition states, and/or validate the estimated production and clearance rates. Additionally, structural connectivity information should be continuously improved, along with available anatomical network reconstruction techniques. The influence on the model variables of other genes identified as essential AB modulators (e.g. BCHE, MGAT3 and CD33 genes [35]-[37], [54]) should be also explored. Since the formulated model (Equation 1) presents the same mathematical structure as the well described predator-prey Lotka-Volterra systems [86], [87], it would be interesting to analyze the individual anatomical stability conditions supporting the intra-brain propagation of the MP factors. Finally, and based on the fact that MP are not the unique factors associated to neurodegenerative progression, more advanced models should be directed towards characterizing MP effects in conjunction with other pathological mechanisms, such as metabolic, vascular and functional dysregulations.

Methods

Ethics statement

The study was conducted according to Good Clinical Practice guidelines, the Declaration of Helsinki, US 21CFR Part 50 – Protection of Human Subjects, and Part 56 – Institutional Review Boards, and pursuant to state and federal HIPAA regulations [88]. Study subjects and/or authorized representatives gave written informed consent at the time of enrollment for sample collection and completed questionnaires approved by each participating sites Institutional Review Board (IRB) [88]. The authors obtained approval from the ADNI Data Sharing and Publications Committee for data use [89] and publication [90].

Intra-brain Epidemic spreading model (ESM) of misfolded proteins (MP) propagation/deposition

Here we consider the brain as a system with N structurally interconnected gray matter regions, where each region i (i = 1..N) is characterized by its temporal probability (P_i) of MP burden. The dynamic behaviour of this system, in terms of MP propagation and deposition, will depend on the interactions between the MP "infested" and "non-infested" regions, where temporal changes in the regional P_i values can be described by the non-linear differential model:

$$\frac{dP_i}{dt} = (1 - P_i(t))\varepsilon_i(t) - \delta_i(t)P_i(t) + \aleph.$$
(1)

The first term on the right side of Equation (1) represents the regional probability of receiving MP infectious-like agents $(\varepsilon_i(t))$ if region *i* is "non-infested" (which happens with probability

 $1 - P_i(t)$). The second term, corresponds to the probability of being clean of MP at time t ($\delta_i(t)$) if region *i* was "infested" before t (which happens with probability $P_i(t)$). The last term, represents an additive noise (\aleph) due to possible stochastic processes, such as natural stochastic factors mediating MP aggregation mechanisms [91] or unknown effects of therapeutic medications.

In the traditional epidemic disease spreading framework, self infection processes are not considered (i.e. the $\varepsilon_i(t)$ term only quantifies the possibility of receiving infectious-like agents from other system entities/nodes). However, in our case, each system entity corresponds to a macroscopic brain region, which comprises several neuronal groups (a direct consequence of the limited PET/MRI spatial resolution and available gray matter parcellation schemes, e.g., [42], [92]). Then, we need to consider the fact that a particular "infested" sub-region in *i* can potentially "infect" neighboring sub-regions. Therefore, $\varepsilon_i(t)$ is modeled as the probabilistic accumulation of exogenous and endogenous infectious-like factors:

$$\varepsilon_i(t) = \sum_{j \neq i} Pa_{j \to i} \beta_j^{ext} (t - \tau_{ij}) P_j (t - \tau_{ij}) + Pa_{i \to i} \beta_i^{int}(t) P_i(t), \quad (2)$$

where $Pa_{j \to i}$ is the weighted anatomical connection probability between the regions j and i (see Anatomical connection probability subsection), $\beta_j^{ext}(t-\tau_{ij})$ is the extrinsic "infection" rate of region jat time $t-\tau_{ij}$, τ_{ij} is the delay corresponding to the time the soluble MP takes to depart from j, with propagation velocity V_{MP} [20], and cover the connection distance $L_{j \to i}$; $\beta_i^{int}(t)$ is the intrinsic "infection" rate of region i at t.

The distinction between extrinsic and intrinsic regional "infection" rates reflects the fact that the total soluble MP produced at a given region is subdivided in two competing processes: the diffusion towards the region's external space (contributing to the global MP expansion) and the molecules staying inside the region (contributing to the production of new seeds and participating in local aggregation mechanisms). As mentioned, soluble MP diffuse from regions of higher concentration to regions of lower concentration. Thus, a high inequality in the deposition levels of all the considered gray matter regions will cause an increase in the extrinsic propagation of soluble MP across the entire brain, and a decrease in the intrinsic fraction of soluble MP that stays in each seed region. These effects are characterized by the relations:

$$\beta_i^{ext}(t) = g(t)\beta_i(t),$$

$$\beta_i^{int}(t) = (1 - g(t))\beta_i(t),$$
(3)

where g is a global tuning variable that quantifies the temporal MP deposition inequality among the different brain regions, and $\beta_i = \beta_i^{ext} + \beta_i^{int}$ is the total "infection" rate of the region *i*. We assume g as the *Gini* coefficient [93], a well established measure of statistical dispersion in a given system, with value 0 reflecting perfect equality and value 1 corresponding to a complete inequality. β_i is defined as a sigmoid function of the regional MP deposition probability, with a high MP deposition probability implying a high probability of producing new infectious-like factors (according to the seeding/nucleation mechanisms, see [94], [95]):

$$\beta_i(t) = \beta_i(P_i, \beta_o) = 1 - e^{-\beta_o P_i(t)}, \tag{4}$$

being $\beta_o \in [0, +\infty]$ an unknown constant parameter.

Similarly to $\beta_i(t)$, the regional probability of being clean of MP after "infection" ($\delta_i(t)$) is expressed as a function of $P_i(t)$ and a constant parameter. However, because MP deposition implies inflammation processes and cellular death, the regional capacity to clear/remove infectious-like agents will decrease with the increase in MP deposition, following a decreasing exponential relationship:

$$\delta_i(t) = \delta_i(P_i, \delta_o) = e^{-\delta_o P_i(t)},\tag{5}$$

where $\delta_o \in [0, +\infty]$ is also an unknown constant parameter. We hypothesize that β_o and δ_o will depend on the specific MP under study, as well as on the individual characteristics (e.g. genetic properties, life style, environmental conditions).

The additive noise (\aleph) in Equation 1 is assumed to follow a Gaussian distribution with unknown mean μ and standard deviation σ . From equations (1)–(5) we see that the proposed model depends on four unknown parameters (β_o , δ_o , μ and σ), which will control the continuous competition between the MP infectious-like agents, the system's clearance response, and the unidentified random processes or external inputs.

Data description and processing

Study participants. Dataset 1: This study used 733 individual data from the Alzheimer's Disease Neuroimaging Initiative (ADNI) [88]. The ADNI was launched in 2003 by the National Institute on Aging (NIA), the National Institute of Biomedical Imaging and Bioengineering (NIBIB), the Food and Drug Administration (FDA), private pharmaceutical companies and non-profit organizations, as a \$60 million, 5-year public-private partnership. The primary goal of ADNI has been to test whether serial magnetic resonance imaging (MRI), positron emission tomography (PET), other biological markers, and clinical and neuropsychological assessment can be combined to measure the progression of mild cognitive impairment (MCI) and early Alzheimer's disease (AD). Determination of sensitive and specific markers of very early AD progression is intended to aid researchers and clinicians to develop new treatments and monitor their effectiveness, as well as lessen the time and cost of clinical trials. The Principal Investigator of this initiative is Michael W. Weiner, MD, VA Medical Center and University of California -San Francisco. ADNI is the result of efforts of many coinvestigators from a broad range of academic institutions and private corporations, and subjects have been recruited from over 50 sites across the U.S. and Canada. The initial goal of ADNI was to recruit 800 subjects but ADNI has been followed by ADNI-GO and ADNI-2. To date these three protocols have recruited over 1500 adults, ages 55 to 90, to participate in the research, consisting of cognitively normal older individuals, people with early or late MCI, and people with early AD. The follow up duration of each group is specified in the protocols for ADNI-1, ADNI-2 and ADNI-GO. Subjects originally recruited for ADNI-1 and ADNI-GO had the option to be followed in ADNI-2 [88]. Written informed consent was obtained from all participants before protocol-specific procedures were performed [88]. For up-to-date information, see www.adni-info.org.

See Table S1 for demographic and clinical characteristics of the included ADNI subjects.

Dataset 2: In addition, this study used the data of 60 young healthy subjects, from the CMU-60 DSI Template (http://www.psy.cmu.edu/~coaxlab/?page_id=423). The CMU-60 DSI Template is a freely available map of reconstructed fiber orientations from very high angular resolution diffusion MRI data, acquired with a 257-direction diffusion spectrum imaging (DSI) sequence. It

was developed by The Cognitive Axon (CoAx) Lab, in the Center for the Neural Basis of Cognition and Department of Psychology at Carnegie Mellon University. The 60 subjects (29 male and 31 female) were recruited from the local Pittsburgh community and the Army Research Laboratory in Aberdeen Maryland. All subjects were neurologically healthy, with no history of either head trauma or neurological or psychiatric illness. Subject ages ranged from 18 to 45 years of age (mean 26 ± 6) at the time of scanning. All acquisition and preprocessing steps described below were performed originally by the members of the CoAx Lab.

Image acquisition. Dataset 1: A 370 MBq (10 mCi+/-10%) bolus injection of AV-45 was administered to each participant, and 20 minute continuous brain PET imaging were acquired approximately 50 minutes post-injection. The images were reconstructed immediately after the 20 minute scan, and when motion artifact was detected, another 20 minute continuous scan was acquired.

Dataset 2: Participants were scanned on a Siemen's Verio 3T system in the Scientific Imaging & Brain Research (SIBR) Center at Carnegie Mellon University using a 32-channel head coil. 50 min, 257-direction DSI scan using a twice-refocused spin-echo EPI sequence and multiple q values (TR = 9,916 ms, TE = 157 ms, voxel size = $2.4 \times 2.4 \times 2.4$ mm, FoV = 231×231 mm, b-max = $5,000 \text{ s/mm}^2$, 51 slices) were collected. Head-movement was minimized during the image acquisition through padding supports and all subjects were confirmed to have minimal head movement during the scan prior to inclusion in the template.

Image processing. Dataset 1: For each individual PET acquisition, images were initially preprocessed according to four main steps [96]: 1) dynamic co-registration (separate frames were co-registered to one another lessening the effects of patient motion), 2) across time averaging, 3) re-sampling and reorientation from native space to a standard voxel image grid space ("AC-PC" space), and 4) spatial filtering to produce images of a uniform isotropic resolution of 8 mm FWHM. Next, all images were spatially normalized to the MNI space [97].

Dataset 2: All images were processed using a q-space diffeomorphic reconstruction method described previously [98]. This method uses a non-linear coregistration approach (ICBM-152 space template regularization, 16 non-linear iterations) that registers the voxel-coordinate into MNI space while also maintaining distortion of the q-space vector during the normalization process. From here, orientation distribution functions (ODFs) were reconstructed to spatial resolution of 2 mm³. The final template image was created by averaging the ODF maps across all 60 subjects, constituting a detailed and unbiased representative map of the nervous fiber orientations in the young healthy brain.

Regional AB deposition patterns. Considering the Cerebellum as an AB non-specific binding reference, individual AB deposition probabilities for 78 regions covering all the brain's gray matter [42] were calculated. First, a bootstrap sampling, consisting of 40,000 randomized sub-samples of the Cerebellum's PET signal values, was used to approximate the extreme value distribution for the maximum values at this region (denoted as PET_{CB}^{maxs}). Next, the maximum likelihood parameters associated to the empirical PET_{CB}^{maxs} distribution were used to calculate the cumulative distribution value of each brain voxel r with signal intensity PET_r . Note that this value will be expressing the probability of the presence, at voxel r, of an equal or higher value than all the possible maximum values within the Cerebellum, i.e. $P(PET_{CB}^{maxs} \leq PET_r)$. Then, the AB deposition probability for a given region $i(P_i^{AB})$ was calculated as:

$$\mathbf{P}_{i}^{\mathrm{AB}} = \frac{1}{|V_{i}|} \sum_{r \in V_{i}} P\left(PET_{CB}^{maxs} \le PET_{r}\right),\tag{6}$$

where V_i is the set of voxels demarked by the region *i*. Finally, these \mathbf{P}_i^{AB} values (i = 1..78) were assumed to represent the individual AB deposition patterns.

Anatomical connection probability. Probabilistic axonal connectivity values between each brain voxel and the surface of each considered gray matter region (voxel-region connectivity) were estimated using a fully automated fiber tractography algorithm [83] and the intravoxel fiber ODFs of the CMU-60 DSI Template. A maximum of 500 mm trace length and a curvature threshold of ± 90 were imposed as tracking parameters.

Based on the resulting voxel-region connectivity maps, the anatomical connection probability between any pair of regions i and j ($ACP_{ij} \equiv ACP_{ji}$) was calculated as the maximum voxel-region connectivity value between both regions. The ACP measure [83] reflects the degree of evidence supporting the existence of each hypothetical white matter connection, independently of the density/strength of this connection, and is thus a measure of low susceptibility to gross fiber degeneration effects related to aging processes. Self connections were considered with $ACP_{ii} = 1$.

Effective anatomical distances to the outbreak regions were estimated as the length of the shortest path (in terms of ACP values) linking region i with the posterior and anterior cingulate cortices [49].

CSF measures. In addition to the neuroimaging data, CSF $A\beta^{1.42}$, t-tau and p-tau¹⁸¹ measurements were acquired for 307 subjects belonging to *Dataset 1*. This subsample comprised HC (n = 78), EMCI (n = 150), LMCI (n = 58) and AD (n = 21) subjects. The xMAP Luminex platform and Innogenetics/Fujirebio AlzBio3 immunoassay kits were used following the SOP in place at the UPenn/ADNI Biomarker Laboratory [68], [99], [100]. Further details on data collection can be found at http://www. adni-info.org. Data were preprocessed as described in [68], [100].

Model exploration/validation with simulated data. Finally, we tried to reproduce the individual PET-based Aß deposition patterns from remote "non-infectious" states. All brain regions, or their combinations up to a maximum of 6 regions (i.e. a total of $78!/(6!\cdot(78-6)!) = 256851595$ combinations), were considered as possible candidates to start the Aß propagation. For each set of sAB spreading seed regions, multiple lifetime trajectories of AB propagation were simulated (Equation 1). Each simulated trajectory consisted of 50 continuous years of hypothetical AB deposition patterns, with each pattern corresponding to a 1 day period. Similar to the creation of the AB reference patterns, each simulated Aß time point pattern (representing one day across the 50 years) consisted of 78 regional probability values, reflecting the local AB "infection" levels. For each subject i (i = [1,733]), we explored iteratively across the parameters space $(\beta_o, \delta_o, \sigma \in \mathcal{R})$, assuming positive values) and the corresponding 50 years trajectories until we identified the set $[\beta_o(i), \delta_o(i), \sigma(i)]$ that minimized, at a unique one day time point t(i), the Euclidean distance between the reference and the simulated AB deposition patterns. Note that, ideally, this time point t(i) should match the ¹⁸F-AV-45 PET scan acquisition day for subject i (Figure 1).

Once we selected the most likely "infectious" seed regions that best explained the reference AB deposition patterns across the study cohort, as well as the optimized individual parameters $[\beta_o(i), \delta_o(i), \sigma(i)]$ and corresponding t(i), we marginalized across all possible regional P values to obtain the individualized global AB production and clearance rates: $\beta_i = \frac{1}{0} \beta(P, \beta_o(i)) dP$ and $\delta_i = \frac{1}{0} \delta(P, \delta_o(i)) dP$, respectively. The individual onset ages of AB binding were also calculated as the difference between the chronological ages and the optimum time points t(i). In sum, for each subject, the individual AB propagation/deposition history was characterized by the set of model parameters $\theta_i = [\beta_i, \delta_i, \sigma_i]$ and the associated onset age (Figure 1).

Statistical analysis

Model cross-validation analysis (Results, first subsection, Figure 2C). Repeated random sub-sampling was used to split each clinical group dataset into training and test data (both data with same sample sizes, n/2). For each split, the model was fitted to the training data, and predictive accuracy (explained variance) was assessed using the test data.

Model variables impact on clinical states (Results, third subsection, Figure 4A). The LMG metric [101] was used, in combination with a Multinomial Logistic Regression, to assess relative importance of regressors. LMG quantify the proportionate contribution of each regressor to the global coefficient of determination (see Grömping, 2006, for review on relative importance metrics). Bootstrapping was used to construct the sampling distributions of the LMG measures and the associated empirical confidence intervals [102].

APOE E4, demographic variables and model-based variables (Results, fourth subsection, Figure 5). Sevenway ANOVA was used to assess predictors association with each model variable. Pair-wise predictor interactions were considered.

Model-based and demographic variables association with CSF measures (Results, final subsection, Figure 6): Seven-way ANOVA was used to assess predictors association with each CSF measure. One-tailed Student's test was used for all the between group

comparisons in the study.

Supporting Information

 $Text\ S1$ Analyzing the ratio between global AB production and clearance rates.

(DOCX)

Text S2 Complete list of ADNI investigators. (PDF)

Figure S1 Relations between effective anatomical distances to all brain regions and Aβ deposition levels. A) Correlations values conserving the original order of the regions in the atlas. B) Absolute correlations after sort the regions from maximum to minimum values, in order to illustrate their natural order as potential propagation seeds. In A) and B), abbreviations are: ACC as anterior cinculate cortex, and PCC as posterior cingulated cortex. (TIF)

Figure S2 Regional A β deposition probability for the different clinical groups vs effective anatomical distance to the outbreak regions. (TIF)

Figure S3 Regional Aß arriving times vs anatomical connectivity degrees, for different Aß probability thresholds (i.e. 0.1, 0.5 and 0.9). Seed regions were not included. (TIF)

Figure S4 Mean (\pm standard error) ratio between Aß production and clearance rates for the different clinical groups (adjusted for gender and educational level). *p<0.05, **p<10⁻⁴, One-tailed Student's *t*-test. (TIF)

 Table S1
 Demographic and clinical characteristics of included

 ADNI subjects.
 (DOCX)

Table S2 Different models explaining Aβ deposition or cortical atrophy associated to AD. (DOCX)

1. 82 Engrander of considered

Table S3 Examples of considered seed regions for starting ABpropagation.(DOCX)

 Table S4
 Contribution of Human Anatomical Connectivity information on MP modeling.

 (DOCX)
 (DOCX)

Table S5Prediction accuracy values obtained with the ESM viaa repeated random sub-sampling cross-validation.(DOCX)

Table S6 Contributions of Effective Anatomical Distance and Spatial Proximity to identified Epicenters on regional Aβ levels. (DOCX)

Table S7 Clinical diagnosis explained by model variables (LMG metric results, after adjusting for gender and educational level).

 (DOCX)

Table S8Model variables differences between clinical groups (t-
test results, after adjusting for gender and educational level).(DOCX)

Table S9Model variables explained by APOE e4 genotype,gender and educational level (ANOVA results).(DOCX)

Table S10 Model variables differences between APOE e4 genotype groups (*t*-test results, after adjusting for gender and educational level). (DOCX)

Table S11 Model variables differences between genders (*t*-test results, after adjusting for APOE e4 genotype and educational level).

(DOCX)

Table S12CSF measures explained by model variables, gender,age and educational level (ANOVA results).(DOCX)

Acknowledgments

We are grateful to Dr. Barry Bedell and Dr. Alejandro Pérez Fernández for their helpful suggestions and comments on the manuscript. We would also like to thank Dr. Tim Verstynen, who kindly collaborated on the use of the CMU-60 DSI Template.

Data used in preparation of this article were obtained from the Alzheimer's Disease Neuroimaging Initiative (ADNI) database [88]. As such, the investigators within the ADNI contributed to the design and implementation of ADNI and/or provided data but did not participate in analysis or writing of this report [89], [90]. A complete listing of ADNI investigators can be found in Text S2.

Author Contributions

Conceived and designed the experiments: YIM RCS. Performed the experiments: YIM. Analyzed the data: YIM. Contributed reagents/ materials/analysis tools: YIM ADNI. Wrote the paper: YIM RCS. Study Supervision: ACE Manuscript revision and associated discussions: YIM RCS PJT ACE.

References

- Dobson CM (2002) Protein misfolding diseases: Getting out of shape. Nature 418: 729–730.
- 2. Dobson CM (2003) Protein folding and misfolding. Nature 426: 884-890.
- 3. Reynaud E (2010) Protein Misfolding and Degenerative Diseases. Nat Educ 3(9): 28.
- Frost B, Diamond MI (2010) Prion-like mechanisms in neurodegenerative diseases. Nat Rev Neurosci 11: 155–159.
- Brundin P, Melki R, Kopito R (2010) Prion-like transmission of protein aggregates in neurodegenerative diseases. Nat Rev 11: 301–307.
- Frost B., Jacks, R. & Diamond M (2009) Propagation of tau misfolding from the outside to the inside of a cell. J Biol Chem 284: 12845–12852.
- Jarrett JT, Berger EP, Lansbury PT (1993) The Carboxy Terminus of the Amyloid Protein Is Critical for the Seeding of Amyloid Formation : Implications for the Pathogenesis of Alzheimer's Disease ? Am Chem Soc 32: 4693–4697.
- Moreno-gonzalez I, Soto C (2011) Seminars in Cell & Developmental Biology Misfolded protein aggregates : Mechanisms, structures and potential for disease transmission. Semin Cell Dev Biol 22: 482–487.
- Collins SR, Douglass A, Vale RD WJ (2004) Mechanism of Prion Propagation: Amyloid Growth Occurs by Monomer Addition. PLoS Biol 2: e321.
- Hallbeck M, Nath S, Marcusson J (2013) Neuron-to-neuron transmission of neurodegenerative pathology. Neuroscientist 19: 560–566.
- Nath S, Agholme L, Kurudenkandy FR, Marcusson J (2012) Spreading of Neurodegenerative Pathology via Neuron-to-Neuron Transmission of beta-Amyloid and Martin Hallbeck. J Neurosci 32: 8767–8777.
- Haass C, Selkoe DJ (2007) Soluble protein oligomers in neurodegeneration: lessons from the Alzheimer's amyloid beta-peptide. Nat Rev Mol Cell Biol 8: 101–112.
- 13. Ittner LM, Jürgen G (2011) Amyloid- β and tau a toxic pas de deux in Alzheimer's disease. Nat Rev Neurosci 12, 67–72.
- Lloret A, Badia M-C, Giraldo E, Ermak G, Alonso M-D, et al. (2011) Amyloidβ toxicity and tau hyperphosphorylation are linked via RCAN1 in Alzheimer's disease. J Alzheimers Dis 27: 701–709.
- Giraldo E, Lloret A, Fuchsberger T, Viña J (2014) Aβ and tau toxicities in Alzheimer's are linked via oxidative stress. Redox Biol 2: 873–877.
- Shankar GM, Li S, Mehta TM, Al E (2008) Amyloid-β protein dimers drom AD impair synaptic plasticity and memory. Nat Med 14, 8, 837–842.
- Desrumaux C, Pisoni A, Meunier J, Deckert V, Athias A, et al. (2013) Increased amyloid-β peptide-induced memory deficits in phospholipid transfer protein (PLTP) gene knockout mice. Neuropsychopharmacology 38: 817–825.
- Lloret A, Badía M-C, Mora NJ, Ortega A, Pallardó F V, et al. (2008) Gender and age-dependent differences in the mitochondrial apoptogenic pathway in Alzheimer's disease. Free Radic Biol Med 44: 2019–2025.
- Song H-L, Shim S, Kim D-H, Won S-H, Joo S, et al. (2014) β-Amyloid is transmitted via neuronal connections along axonal membranes. Ann Neurol 75: 88–97.
- Waters J (2010) The concentration of soluble extracellular amyloid-β protein in acute brain slices from CRND8 mice. PLoS One 5: e15709.
- Jucker M, Walker LC (2013) Self-propagation of pathogenic protein aggregates in neurodegenerative diseases. Nature 501: 45–51.
- Walker LC, LeVine H (2012) Corruption and spread of pathogenic proteins in neurodegenerative diseases. J Biol Chem 287: 33109–33115.
- Braak H BE (1991) Neuropathological stageing of Alzheimer-related changes. Acta Neuropathol 82: 239–259.
- Braak H, Ghebremedhin E, Rub U, Bratzke H DTK (2004) Stages in the development of Parkinson's disease-related pathology. Cell Tissue Res 318: 121–134.
- DiFiglia M, Sapp E, Chase KO, Davies SW, Bates GP, Vonsattel JP AN (1997) Aggregation of huntingtin in neuronal intranuclear inclusions and dystrophic neurites in brain. Science 277: 1990–1993.
- Palop JJ, Chin J, Mucke L (2006) A network dysfunction perspective on neurodegenerative diseases. Nature 443: 768–773.
- Seeley WW, Crawford RK, Zhou J, Miller BL, Greicius MD (2009) Neurodegenerative Diseases Target Large-scale Human Brain Networks. Neuron 62: 42–52.
- Greicius MD, Srivastava G, Reiss AL M V. (2004) Default-mode network activity distinguishes Alzheimer's disease from healthy aging: evidence from functional MRI. Proc Natl Acad Sci USA 101: 4637–4642.
- Buckner R, Snyder A, Shannon B, LaRossa G, Sachs R, et al. (2005) Molecular, structural, and functional characterization of Alzheimer's disease: evidence for a relationship between default activity, amyloid, and memory. J Neurosci 25: 7709–7717.
- Zhou J, Gennatas ED, Kramer JH, Miller BL, Secley WW (2012) Predicting Regional Neurodegeneration from the Healthy Brain Functional Connectome. Neuron 73: 1216–1227.
- Buckner RL, Sepulcre J, Talukdar T, Krienen FM, Liu H, et al. (2009) Cortical hubs revealed by intrinsic functional connectivity: mapping, assessment of stability, and relation to Alzheimer's disease. J Neurosci 29: 1860–1873.
- De Haan W, Mott K, van Straaten ECW, Scheltens P, Stam CJ (2012) Activity dependent degeneration explains hub vulnerability in Alzheimer's disease. PLoS Comput Biol 8: e1002582.

- Stam CJ, Haan W De, Daffertshofer A, Jones BF, Manshanden I, et al. (2009) Graph theoretical analysis of magnetoencephalographic functional connectivity in Alzheimer's disease. Brain 132(1): 213–24.
- Raj A, Kuceyeski A, Weiner M (2012) A Network Diffusion Model of Disease Progression in Dementia. Neuron 73: 1204–1215.
- 35. Fiala M, Mahanian M, M R, Mizwicki M, Tse E, et al. (2011) MGAT3 mRNA: a biomarker for prognosis and therapy of Alzheimer's disease by vitamin D and curcuminoids. J Alzheimers Dis 25: 135–144.
- Cashman J, Ghirmai S, Abel K, Fiala M (2008) Immune defects in Alzheimer's disease: new medications development. BMC Neurosci 9(2): S13.
- Griciuc A, Serrano-Pozo A, Parrado A, Lesinski A, Asselin C, et al. (2013) Alzheimer's disease risk gene CD33 inhibits microglial uptake of amyloid beta. Neuron 78: 631–643.
- Mawuenyega K, Sigurdson W, Ovod V, Munsell L, Kasten T, et al. (2010) Decreased clearance of CNS beta-amyloid in Alzheimer's disease. Science 330(6012): 1774.
- Domert J, Rao SB, Agholme L, Brorsson A-C, Marcusson J, et al. (2014) Spreading of amyloid-β peptides via neuritic cell-to-cell transfer is dependent on insufficient cellular clearance. Neurobiol Dis 65: 82–92.
- Soscia SJ, Kirby JE, Washicosky KJ, Tucker SM, Ingelsson M, et al. (2010) The Alzheimer's disease-associated amyloid beta-protein is an antimicrobial peptide. PLoS One 5: e9505.
- Grant J, Ghosn E, Axtell R, Herges K, Kuipers H, et al. (2012) Reversal of paralysis and reduced inflammation from peripheral administration of βamyloid in TH1 and TH17 versions of experimental autoimmune encephalomyelitis. Sci Transl Med 4(145): 145ra105.
- Klein A, Tourville J (2012) 101 Labeled Brain Images and a Consistent Human Cortical Labeling Protocol. Front Neurosci 6: 171.
- Buckner R, Andrews-Hanna J, Schacter D (2008) The brain's default network: anatomy, function, and relevance to disease. Ann N Y Acad Sci 1124: 1–38.
- Andrews-Hanna J, Reidler J, Huang C, Buckner R (2010) Evidence for the default network's role in spontaneous cognition. J Neurophysiol 104(1): 322– 335.
- Mevel K, Chételat G, Eustache F, Desgranges B (2011) The Default Mode Network in Healthy Aging and Alzheimer's Disease. Int J Alzheimers Dis, Article ID 535816.
- Leech R, Sharp D (2013) The role of the posterior cingulate cortex in cognition and disease. Brain 137(1): 12–32.
- Rubinov M, Sporns O (2011) Weight-conserving characterization of complex functional brain networks. Neuroimage 56: 2068–2079.
- Snow J (1855) On the mode of communication of cholera. London J Churchill, 2.
- Brockmann D, Helbing D (2013) The hidden geometry of complex, networkdriven contagion phenomena. Science 342: 1337–1342.
- Van den Heuvel MP, Sporns O (2013) Network hubs in the human brain. Trends Cogn Sci 17: 683–696.
- Crossley N a., Mechelli a., Scott J, Carletti F, Fox PT, et al. (2014) The hubs of the human connectome are generally implicated in the anatomy of brain disorders. Brain 137(8): 2382–2395.
- Hardy J, Higgins G (1992) Alzheimer's disease: the amyloid cascade hypothesis. Science 256: 184–185.
- Karran E, Mercken M, De Strooper B (2011) The amyloid cascade hypothesis for Alzheimer's disease: an appraisal for the development of therapeutics. Nat Rev Drug Discov 10: 698–712.
- Ramanan V, Risacher S, Nho K, Kim S, Swaminathan S, et al. (2013) APOE and BCHE as modulators of cerebral amyloid deposition: a florbetapir PET genome-wide association study. Mol Psychiatry 19, 351–357.
- Holtzman D, Morris J, Goate A (2011) Alzheimer's disease: the challenge of the second century. Sci Transl Med 3: 77sr71.
- Castellano J, Kim J, Stewart F, Jiang H, DeMattos R, et al. (2011) Human apoE isoforms differentially regulate brain amyloid-beta peptide clearance. Sci Transl Med 3: 89ra57.
- Koffie R, Hashimoto T, Tai H, Kay K, Serrano-Pozo A, et al. (2012) Apolipoprotein E4 effects in Alzheimer's disease are mediated by synaptotoxic oligomeric amyloid-beta. Brain 135: 2155–2168.
- Corder E, Saunders A, Strittmatter W, Schmechel D, Gaskell P, et al. (1993) Gene dose of apolipoprotein E type 4 allele and the risk of Alzheimer's disease in late onset families. Science (80) 261: 921–923.
- Mosconi L, Herholz K, Prohovnik I, Nacmias B, De Cristofaro M, et al. (2005) Metabolic interaction between ApoE genotype and onset age in Alzheimer's disease: implications for brain reserve. J Neurol Neurosurg Psychiatry 76(1): 15–23.
- Genin E, Hannequin D, Wallon D, Sleegers K, Al E (2011) APOE and Alzheimer disease: a major gene with semi-dominant inheritance. Mol Psychiatry 16(9): 903–907.
- Damoiseaux J, Seeley W, Zhou J, Shirer W, Coppola G, et al. (2012) Gender Modulates the APOE E4 Effect in Healthy Older Adults: Convergent Evidence from Functional Brain Connectivity and Spinal Fluid Tau Levels. J Neurosci 32(24): 8254–8262.
- 62. Farrer L, Cupples L, Haines J, Hyman B, Kukull W, et al. (1997) Effects of age, sex, and ethnicity on the association between apolipoprotein E genotype and

Alzheimer disease: a meta-analysis. APOE and Alzheimer Disease Meta Analysis Consortium. JAMA 278: 1349-1356.

- 63. Corder E, Ghebremedhin E, Taylor M, Thal D, Ohm T, et al. (2004) The biphasic relationship between regional brain senile plaque and neurofibrillary tangle distributions: modification by age, sex, and APOE polymorphism. Ann N Y Acad Sci 1019: 24-28.
- 64. Rodrigue K, Kennedy K, Park D (2009) Beta-Amyloid Deposition and the Aging Brain. Neuropsychol Rev 19(4): 436–450.
- 65. Aizenstein H, Nebes R, Saxton J, Price J, Mathis C, et al. (2008) Frequent amyloid deposition without significant cognitive impairment among the elderly. Arch Neurol 65: 1509–1517.
- 66. Kemppainen N, Aalto S, Karrasch M, Någren K, Savisto N, et al. (2008) Cognitive reserve hypothesis: Pittsburgh Compound B and fluorodeoxyglucose positron emission tomography in relation to education in mild Alzheimer's disease. Ann Neurol 63: 112-118.
- 67. Rolstad S, Nordlund A, Eckerström C, Gustavsson M, Zetterberg H, et al. (2009) Biomarkers in relation to cognitive reserve in patients with mild cognitive impairment-proof of concept. Dement Geriatr Cogn Disord 27: 194-200
- Shaw LM, Vanderstichele H, Knapik-Czajka M et al. (2009) Cerebrospinal 68 fluid biomarker signature in Alzheimer's disease neuroimaging initiative subjects. Ann Neurol 65: 403-413.
- Jack CR Jr, Knopman DS, Jagust WJ et al. (2010) Hypothetical model of 69 dynamic biomarkers of the Alzheimer's pathological cascade. Lancet Neurol 9: 119 - 128
- 70. Jack CR, Knopman DS, Jagust WJ, Petersen RC, Weiner MW, et al. (2013) Tracking pathophysiological processes in Alzheimer's disease: an updated hypothetical model of dynamic biomarkers. Lancet Neurol 12: 207-216.
- 71. Small S, Duff K (2008) Abeta and tau in late-onset Alzheimer's disease: a dual pathway hypothesis. Neuron 60: 534-542.
- 72. Metcalf DJ, García-Arencibia M, Hochfeld WE, Rubinsztein DC (2012) Autophagy and misfolded proteins in neurodegeneration. Exp Neurol 238: 22-28
- 73. Palop J, Mucke L (2010) Amyloid-B Induced Neuronal Dysfunction in Alzheimer's Disease: From Synapses toward Neural Networks. Nat Neurosci 13(7): 812-818.
- 74. Estrada E, Kalala-mutombo F, Valverde-colmeiro A (2011) Epidemic spreading in networks with nonrandom long-range interactions. Phys Rev E 036110: 1-15.
- 75. Iturria-medina Y (2013) Anatomical Brain Networks on the Prediction of Abnormal Brain States. Brain Connectivity 3(1): 1-21.
- 76. Evans A (2013) Networks of anatomical covariance. Neuroimage 80: 489-504.
- 77. Rinne JO, Brooks DJ, Rossor MN, Fox NC, Bullock R, et al. (2010) 11C-PiB PET assessment of change in fibrillar amyloid-B load in patients with Alzheimer's disease treated with bapineuzumab: a phase 2, double-blind,
- placebo-controlled, ascending-dose study. Lancet Neurol 9(4): 363-372. 78. Potter R, Patterson BW, Elbert DL, Ovod V, Kasten T, et al. (2013) Increased in vivo amyloid-\$42 production, exchange, and loss in presenilin mutation carriers. Sci Transl Med 5: 189ra77.
- Wildsmith K, Holley M, Savage J, Skerrett R, Landreth G (2013) Evidence for 79. impaired amyloid ß clearance in Alzheimer's disease. Alzheimers Res Ther 5(4): 33.
- 80. Wisniewski T, Goñi F (2014) Immunotherapy for Alzheimer's disease. Biochem Pharmacol 88: 499-507.

- 81. Cramer P, Cirrito J, Wesson D, Lee C, Karlo J, et al. (2012) ApoE-directed therapeutics rapidly clear β-amyloid and reverse deficits in AD mouse models. Science 335(6075): 1503-1506.
- 82. Jones DK, Cercignani M (2010) Twenty-five pitfalls in the analysis of diffusion MRI data. NMR Biomed 23: 803-820
- 83. Iturria-Medina Y, Canales-Rodríguez EJ, Melie-García L, Valdés-Hernández PA, Martínez-Montes E, et al. (2007) Characterizing brain anatomical connections using diffusion weighted MRI and graph theory. Neuroimage 36: 645-660.
- 84. Zalesky A, Fornito A, Harding IH, Cocchi L, Yücel M, et al. (2010) Wholebrain anatomical networks: Does the choice of nodes matter? Neuroimage 50: 970-983
- 85. Caspers S, Eickhoff SB, Zilles K, Amunts K (2013) Microstructural grey matter parcellation and its relevance for connectome analyses. Neuroimage 80: 18-26.
- 86. Lotka A (1925) Elements of physical biology. Williams Wilkins, Balt
- 87. Volterra V (1926) Variazioni e fluttuazioni del numero d'individui in specie animali conviventi. Lincei, Mem R Accad Naz dei 2, 31-113.
- 88. http://adni.loni.usc.edu.
- 89. http://adni.loni.usc.edu/wp-content/uploads/how_to_apply/ADNI_Data_ Use_Agreement.pdf. 90. http://adni.loni.usc.edu/wp-content/uploads/how_to_apply/ADNI_
- Manuscript_Citations.pdf. 91. Hortschansky P, Schroeckh V, Christopeit T, Zandomeneghi G, M F (2005)
- The aggregation kinetics of Alzheimer's beta-amyloid peptide is controlled by stochastic nucleation. Protein Sci 14: 1753-1759.
- Mazziotta J, Toga A, Evans A, Fox P, Lancaster J (1995) A Probablistic Atlas of 92. the Human Brain: Theory and Rationale for Its Development. Neuroimage 2: 89-101
- Gini C (1912) Variability and Mutability. C Cuppini, Bol: 156 pages.
 Lundmark K, Westermark G, Nystrom S, Murphy C, Solomon A, et al. (2002) Transmissibility of systemic a myloidosis by a prion-like mechanism. Proc Natl Acad Sci U S A 99(10): 6979-6984.
- 95. Cui D, Kawano H, Takahashi M, Hoshii Y, Setoguchi M, et al. (2002) Acceleration of murine AA amyloidosis by oral administration of amyloid fibrils extracted from different species. Pathol Int 52(4): 264.
- 96. Jagust W, Bandy D, Chen K, Foster N, Landau S, et al. (2010) The Alzheimer's disease neuroimaging initiative positron emission tomography core. Alzheimers Dement 6: 221-229.
- 97. Evans A, Kamber M, Collins D, Macdonald D (1994) An MRI-based probabilistic atlas of neuroanatomy. S Shorvon, D Fish, F Andermann, G -Bydder H Stefan, Ed Magn Reson Scanning Epilepsy, Plenum, New York: 263 - 274
- 98. Yeh F, Tseng W (2011) NTU-90: a high angular resolution brain atlas constructed by q-space diffeomorphic reconstruction. Neuroimage 58(1): 91-00
- Olsson A, Vanderstichele H, Andreasen N, DeMeyer G, Wallin A, et al. (2005) 99. Simultaneous measurement of B-amyloid1-42 in CSF by xMAP technology. Clin Chem 51: 336-345
- 100. Shaw L, Vanderstichele H, Knapik-Czajka, Figurski M, Coart E, et al. (2011) Qualification of the analytical and clinical performance of CSF biomarker analyses in ADNI. Acta Neuropath 121: 597-609.
- 101. Lindeman R, Merenda P, Gold R (1980) Introduction to Bivariate and Multivariate Analysis. Scott, Foresman, Glenview, IL: 119.
- 102. Grömping U (2006) Relative Importance for Linear Regression in R : The Package relaimpo. J Stat Sofw 17 (1).